Author's response to reviews

Title: The Changing Epidemiology of Dengue in China, 1990-2014: a descriptive analysis of 25 years of Nationwide Surveillance Data

Authors:

Shengjie Lai (laisj@chinacdc.cn)
Zhuojie Huang (seenhzj@gmail.com)
Hang Zhou (zhouhang@chinacdc.cn)
Katherine L Anders (kanders@oucru.org)
T A Perkins (taperkins@gmail.com)
Wenwu Yin (yinww@chinacdc.cn)
Yu Li (liyu1@chinacdc.cn)
Di Mu (mudi@chinacdc.cn)
Qiulan Chen (chenql@chinacdc.cn)
Zike Zhang (zhangzk1@chinacdc.cn)
Yanzi Qiu (qiuyz@chinacdc.cn)
Liping Wang (wanglp@chinacdc.cn)
Honglong Zhang (zhanghl@chinacdc.cn)
Linjia Zeng (zenglj@chinacdc.cn)
Xiang Ren (renxiang@chinacdc.cn)
Mengjie Geng (gengmj@chinacdc.cn)
Zhongjie Li (lizj@chinacdc.cn)
Andrew J Tatem (andy.tatem@gmail.com)
Simon I Hay (simon.i.hay@gmail.com)
Hongjie Yu (yuhj@chinacdc.cn)

Version: 4  Date: 21 March 2015

Author's response to reviews:

Re: Revision of manuscript 1643447191157616

Dear Dr. Barnard,

Thank you for the opportunity to revise our manuscript “The Changing Epidemiology of Dengue in China, 1990-2014: a descriptive analysis of 25 years of Nationwide Surveillance Data”. We have read through the reviewers’ comments and have addressed each issue below. You will find a point-by-point response to each comment in blue, with a highlighted reference to the page and paragraph where changes have been made in the manuscript. We thank the reviewers for their insightful comments and feel the manuscript is much stronger after taking these into account. We look forward to hearing from you and your team.

Warmest regards,
Hongjie Yu, MD, MPH, PhD
Division of Infectious Diseases,
Key Laboratory of Surveillance and Early-warning on Infectious Disease,
Chinese Center for Disease Control and Prevention